In ternational Symp osium on Nuclear Electro-W eak Sp ectroscopy (NEWS 99) for -Symmetries and Electro-W eak Nuclear pro cesses- The international symp osium on Nuclear Electro-W eak Sp ectroscopy (NEWS 99) for Symmetries and Electro-W eak Nuclear pro cesses was held in Icho-Kaik an of Osak a Universit y from March 9 to 12 of 1999. RCNP w as the host institute of this symp osium. Nuclei are systems of n ucleons in go o d quantum states. Th us, they are used as excellen t microlab oratories for studying elementary particles and fundamen tal interactions. Symme- tries and electro-w eak pro cesses are studied by inv estigating electro-w eak nuclear-processes in n uclear microlab oratories. High precision studies by nuclear electro-weak sp ectroscopies are very useful for symmetry studies R CNP (Research Center for Nuclear Ph ysics) Osak a Univ. is a mo dern lab oratory complex. It consists of the K=0.4 GeV Ring Cyclotron lab oratory for nucleon-meson nuclear physics, the Oto Cosmo Observ atory (underground lab oratory) for lepton (neutrinos, dark matters) nuclear ph ysics with high sensitive ELEGANT detectors, and of the Laser Electron Photon lab oratory at SPring-8 with multi-GeV p olarized photons for quark nuclear ph ysics. Symmetries and electro-w eak nuclear-processes are key sub jects. These directions of RCNP had b een promoted strongly by Prof. Hiro Ejiri, the RCNP director, since 1993. Thus, w e organized this symp osium in honor of Prof. Hiro Ejiri, who terminated the third term of the directorship of RCNP and the Professorship of Osak a Universit y in March 31st, 1999. Since the conference on quark lepton n uclear ph ysics was held in May 1997 at RCNP . The present symp osium puts emphasis on symmetries and electro-weak pro cesses. The sub jects discussed w ere: 1). Symmetries in Electro-W eak In teractions in Nuclei 2). Neutrinos and Symmetries in Nuclei 3). Dynamical Symmetry Breaking and QCD Ph ysics 4). Quarks and Hadrons by Electro W eak Prob es 5). Symmetries in Flav or Nuclear Particle Physics 6). Symmetries in Nuclear Structures b y Electromagnetic Sp ectroscopy 7). Astro Nuclear Physics by Electro W eak Nuclear Pro cesses 8). Related Problems W e had ab out 150 participants from v arious countries. The pro ceedings of this con- ference will b e published from W orld Scienti�c. In addition, utilizing this o ccasion, w e had a workshop on the double b eta deca y and the dark matter after NEWS99 at RCNP and the Oto Cosmo Lab oratory . (Mini W orkshop on \Neutrinos and Nuclear Rare Pro cesses in Astroparticle Physics" (OTO99), March 13-14, 1999)

Host Institute: Researc h Cen ter for Nuclear Physics (R CNP), Osak a Universit y International Advisory Committee Organizing Committee J. Arvieux (Orsa y) H. T oki (R CNP) Chair. P . Barnes (Los Alamos) T. Kishimoto (Osak a) Co-Chair. J. Cameron (Indiana) K. Hatanak a (RCNP) Scienti�c Secretary R. Chrien (BNL) J. Deutch (Luvin) M. F ujiw ara (RCNP) A. F asseler (T ubingen) T. F ukuda (KEK) E. Fiorini (Milan) A. Hosak a (Numazu) A. Gal (Jerusalem) Y. Kuno (KEK) M. Harakeh (KVI) K. Maeda (T ohoku) E. Henley (W ashington) T. Motobay ashi (Rikkyo) Ph. Hub ert (Bordeaux) K. Muto (TIT) H. Kamitsub o (SPring-8) Y. Nagai (TIT) T. Kirsten (Max-Plank) T. Nak ano (RCNP) K. Kub o (T oky o Metrop olitan) M. Nomachi (RCNP) S. Kullander (Uppsala) T. Noro (R CNP) V. Lobashev (Mosco w) H. Ohsumi (Saga) A. Masaike (Kyoto) K. Ok ada (Ky oto Sangyo) A. Morales (Zaragoza) H. Sak ai (T oky o) H. Ohtsubo (Osak a) K. Sato (RCNP) A. Sanda (Nago y a) T. Sato (Osak a) O. Sch ult (Julich, IKP) T. Shibata (KEK) Z. Sun (Beijing) K. T ak ahisa (RCNP) A. Suzuki (T ohoku) A. Thomas (Adelaid) W. W eise (Munich) K. Y azaki (T oky o)